Author’s response to reviews

Title: Outcome of MS relapses in the era of disease-modifying therapy

Authors:
Muriel Stoppe (muriel.stoppe@medizin.uni-leipzig.de)
Maria Busch (maria.busch@medizin.uni-leipzig.de)
Luise Krizek (Luise.Krizek@medizin.uni-leipzig.de)
Florian Then Bergh (thenberf@medizin.uni-leipzig.de)

Version: 3 Date: 14 Jul 2017

Author’s response to reviews:

Dear Editor,

thank you for the email message as of Jul 5th, 2017. Again, we appreciate the time and effort put into the manuscript by the reviewers, Drs Monson and Chan, and yourself. We were happy to see that we met both reviewer’s requests, and were also happy to address your essential minor revisions. Please find enclosed a revised (“clean”) version of the manuscript. We would like to list our modifications according to your suggestions.

1 Key words
Authors´ response: Key words were placed after the abstract.

2 Discussion and conclusions
Authors´ response: A separate “Conclusion” section was introduced.
3 Ethics approval and consent to participate

Authors´ response: The sections were combined to a single section and the regulatory situation explained further, adding the reference to the pertinent legal regulation.

4 Acknowledgements and Funding

Authors´ response: The information about funding was moved from “Acknowledgements” to “Funding”, and we added a statement that the public funding source took no role in the design or conduct of the study.

5 Cover Letter

Authors´ response: The previous cover letter was removed from the file inventory.

6 Clean Copy

Authors´ response: Only a clean copy was uploaded.

Again, we thank the reviewers for their input which has certainly contributed to improving our manuscript.

Sincerely,

Prof. Dr. Florian Then Bergh

Consultant Neurologist

Neuroimmunology